
• Part of NuMI Tritium Migigations, CUB off Friday, April 21 through Monday, April 24th.   This 
will be followed by the next site-wide power outage.

Rotate AP2 BPM 

0% done

Install OTR 

0% done

Install New Beam Stop 

40% done

Install AP2 Trims 

100% done

AP0 SO cord replacement

100% done

Install New Target 

0% done

AP2 Vacuum gate 

95% done

Motor. Kicker Stand 

15% done
New D10 Pipe 

100% done

Rotate D/A BPMs 

100% done
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Quad BPM 

90% done

AP30 Tank Leak 

100% done
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Upgrade Deb. BPM 

90% done

Feeder 24 

100% done

Survey Level Run 

85% done

Install Hall Probes 

5% done

Beam Line BPM Cable 

5% done
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Acc BPMs 

0% done

LM Calib. 

50% done

Deb Quad Shunts. 

100% done

4-8GHz Core 

10% done

AP1 PS 

50% done

AP2 BPM pre-amps

10%

• Motorized kicker  - 20% done.   All modules taken apart.  Sandblasting.   Stand is waiting for 
kicker to come back.  

• New trim supply.   Controls problems.   453/463 are similar.   But 463 cant point back to 218 
card.    This is a work in progress, but the problems should be able to be worked out.

• 4-8GHz core - new plates 11nsec faster.   20% done.
• Stacktail prototype - 40%
• AP2 gate valve.   Cable has been pulled, but not sure if it is terminated.
• Debuncher BPMs - 90%
• LM calibratoin - 60%
• AP2 BPM pre-amps - 20%.  Half are built, but none are installed
• Survey level run - may be done?
• New AP2 beam stop - 70%.  What is left is installation of interlock electronics and testing.   

Not yet aligned.
• Rotate AP2 BPM - not started.   
• OTRs -
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• OTRs -
• AP1 power supply - 70%  - has some database issues.
• Hall probes - pull more cables in AP0 to get analog channels
• Beam line BPM cables. 
• Target scheduled in May
• Saturday 12pm to 6pm, severe storm seminar.

Toroid 724.   Top pulse is what we are sending down to the Toroid to simulate beam, the 

bottom is what we get back.
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